Author's response to reviews

Title: Self-rated health among Mayan women participating in a randomised intervention trial reducing indoor air pollution in Guatemala

Authors:

Esperanza Diaz (esperanza.diaz@isf.uib.no)
Nigel Bruce (ngb@liverpool.ac.uk)
Dan Pope (danpope@liverpool.ac.uk)
Anaite Diaz (anaite_artiga@yahoo.com)
Kirk R Smith (krksmith@berkeley.edu)
Tone Smith-Sivertsen (tone.smith-sivertsen@isf.uib.no)

Version: 6 Date: 3 June 2008

Author's response to reviews: see over
Dear Editor:

Thank you very much for the accepting our paper Self-rated health among Mayan women participating in a randomised intervention trial reducing indoor air pollution in Guatemala.

We hereby resubmit the manuscript having made the suggested formatting requirements.

Best wishes

Esperanza Diaz